Author’s response to reviews

Title: The Specific and Combined role of Domestic Violence And Mental Health Disorders During Pregnancy on New-born Health

Authors:

Alexandre Ferraro (ferraro@usp.br)
Luis Rohde (lrohde@terra.com.br)
Guilherme Polanczyk (polanczyk.guilherme@gmail.com)
Adriana Argeu (adrianaargeu@gmail.com)
Euripides Miguel (ecmiguel7@gmail.com)
Sandra Josefina Grisi (sandra.grisi@hc.fm.usp.br)
Bacy Fleitlich-Bilyk (bacy@uol.com.br)

Version: 4 Date: 14 Jul 2017

Author’s response to reviews:

In this final review we respond to what has been requested:

- a manuscript version that has been revised by a native speaker;
- we have completed the Declaration Section with the information that was lacking.

Best regards.